Author's response to reviews

Title: Improving healthcare consumer effectiveness: An Animated, Self-serve, Web-based Research Tool (ANSWER) for people with early rheumatoid arthritis

Authors:

Linda C Li (lli@arthritisresearch.ca)
Paul Adam (Paul.Adam@vch.ca)
Anne F Townsend (atownsen@interchange.ubc.ca)
Dawn Stacey (Dawn.Stacey@uOttawa.ca)
Diane Lacaille (dlacaille@arthritisersearch.ca)
Susan Cox (suecox@interchange.ubc.ca)
Jessie McGowan (jmcgowan@uottawa.ca)
Peter Tugwell (ptugwell@uottawa.ca)
Gerri Sinclair (gerris@gnwc.ca)
Kendall Ho (kendall.ho@ubc.ca)
Catherine L Backman (catherine.backman@ubc.ca)

Version: 5 Date: 14 August 2009

Author's response to reviews:

The link to the CIHR funding decision webpage has been updated.